Author's response to reviews

Title: Development and preliminary validation of a Greek-language outpatient satisfaction questionnaire with principal components and multi-trait analyses.

Authors:

Vassilis H Aletras (valetras@uom.gr)
Efthimis A Papadopoulos (efthimiospapado@hotmail.com)
Dimitris A Niakas (niakas@eap.gr)

Version: 5  Date: 6 June 2006

Author’s response to reviews:

Dear Editors,

As requested, we have made the final changes to the format of our manuscript. Specifically, we have included an Acknowledgement Section and removed the Table placement text. In addition, we have corrected a few remaining errors in the presentation and language.

Yours faithfully,

Vassilis Aletras